\begin{thebibliography}{10}

\bibitem{365931}
S.~G. Tucker.
\newblock Emulation of large systems.
\newblock {\em Commun. ACM}, 8(12):753--761, 1965.

\bibitem{microprogramming.history}
Mark Smotherman.
\newblock A brief history of microprogramming.
\newblock \\http://www.cs.clemson.edu/~mark/uprog.html, 1999.

\bibitem{804781}
J.~C. Demco and T.~A. Marsland.
\newblock An insight into {PDP-11} emulation.
\newblock In {\em MICRO 9: Proceedings of the 9th annual workshop on
  Microprogramming}, pages 20--26, New York, NY, USA, 1976. ACM Press.

\bibitem{transmeta-site}
Transmeta site.
\newblock \\http://www.transmeta.com/.

\bibitem{fhs-site}
Filesystem hierarchy standard.
\newblock \\http://www.pathname.com/fhs/pub/fhs-2.3.html, 01 2004.

\bibitem{debian-policy}
The debian-policy~mailing list.
\newblock Debian policy manual.
\newblock \\http://www.debian.org/doc/debian-policy/.

\bibitem{tis.elf}
{Tool Interface Standard (TIS) Executable and Linking Format (ELF)
  Specification}.
\newblock \\http://www.x86.org/ftp/manuals/tools/elf.pdf, 05 1995.

\bibitem{qemu-site}
Qemu web site.
\newblock \\http://fabrice.bellard.free.fr/qemu/.

\bibitem{vmware-site}
Vmware web site.
\newblock \\http://www.vmware.com/.

\bibitem{DBLP:conf/scm/ChristensenE95}
Anders Christensen and Tor Egge.
\newblock Store - a system for handling third-party applications in a
  heterogeneous computer environment.
\newblock In Jacky Estublier, editor, {\em SCM}, volume 1005 of {\em Lecture
  Notes in Computer Science}, pages 263--276. Springer, 1995.

\bibitem{tfheen-scalable}
Tollef~Fog Heen.
\newblock Scalable software distribution systems.
\newblock Technical report, Norwegian University of Science and Technology,
  Department of Computer and Information Science, 2004.
\newblock \\http://err.no/ntnu/project/report.pdf.

\bibitem{binutils-site}
binutils web site.
\newblock \\http://www.gnu.org/software/binutils/.

\bibitem{gcc-site}
{GCC} web site.
\newblock \\http://gcc.gnu.org/.

\bibitem{pkgconfig-site}
pkgconfig web site.
\newblock \\http://pkgconfig.freedesktop.org.

\bibitem{libtool-site}
libtool web site.
\newblock \\http://www.gnu.org/software/libtool/libtool.

\bibitem{autoconf-site}
autoconf web site.
\newblock \\http://www.gnu.org/software/autoconf/.

\bibitem{automake-site}
automake web site.
\newblock \\http://www.gnu.org/software/automake/.

\bibitem{netbsd-site}
{NetBSD}.
\newblock http://www.netbsd.org/ports/.

\bibitem{freebsd-site}
{FreeBSD web site}.
\newblock http://www.freebsd.org/.

\bibitem{itanium-site}
{Itanium/ia64 web site at Intel}.
\newblock \\http://developer.intel.com/design/itanium/family/.

\bibitem{taggart-multiarch}
Matt Taggart.
\newblock Multiple architecture problem and proposed solution.
\newblock http://www.linuxbase.org/~taggart/multiarch.html, 06 2004.

\bibitem{tfheen-multiarch3}
Tollef~Fog Heen.
\newblock How to handle multiarch on x86-64 (and other platforms).
\newblock http://err.no/debian/amd64-multiarch-3, 05 2004.

\bibitem{tfheen-multiarch2}
Tollef~Fog Heen.
\newblock How to handle multiarch on x86-64 (and other platforms).
\newblock http://err.no/debian/amd64-multiarch-2, 02 2004.

\bibitem{dpkg-site-featuredepends}
Scott~James Remnant.
\newblock Dpkg web site: Feature dependencies.
\newblock \\http://www.dpkg.org/FeatureDependencies.

\bibitem{debian-policy/binarypackage/control}
The debian-policy~mailing list.
\newblock Debian policy manual; binary package control files -- debian/control.
\newblock
  \\http://www.debian.org/doc/debian-policy/ch-controlfields.html\#s-binarycon%
trolfiles.

\bibitem{debian-policy/sourcepackage/control}
The debian-policy~mailing list.
\newblock Debian policy manual; source package control files -- debian/control.
\newblock
  \\http://www.debian.org/doc/debian-policy/ch-controlfields.html\#s-sourcecon%
trolfiles.

\end{thebibliography}

\chapter{Analysis of existing solutions}

The existing solutions have a few common shortcomings:

\begin{itemize}
\item Limited number of architectures
\item No way to handle multiple architectures with same pointer size
\item Little or no real support for multiarch, just hacked-on solutions
\end{itemize}

There are some existing solutions for handling multiarch and there has
been a consensus forming on how to solve it on the file system
hierarchy level, but nobody has so far been willing to take the first
step and make their system multiarch-compliant.  The implementations
which exist are limited and usually only supports two architectures,
not an unlimited number.

In most cases, support for two architectures will be enough.  A very
common case will be support for i386 binaries on an x86\_64 machine or
support for powerpc or sparc binaries on their equivalent 64 bit
systems.  This can be solved adequately with a solution which just
provides support for two architectures, for instance the file system
layout most Linux distributions have so far with \directory{/usr/lib}
and \directory{/usr/lib64} (or \directory{/usr/lib} and
\directory{/usr/lib32} where the ``native'' architecture is 64 bit).

The lib/lib64 solution doesn't work well once you want to support
multiple 64 or 32 bit architectures.  An example is
NetBSD\cite{netbsd-site} and FreeBSD's\cite{freebsd-site} support for
running Linux binaries.  Another example would be the moment
ia64\cite{itanium-site} get support for running x86\_64 binaries where
you'd get a collision between the architectures in what should go in
/usr/lib (since that is a 64 bit directory on ia64).

One can construct more contrived examples, such as running i386 and
amd64 binaries for Linux, FreeBSD and NetBSD, on a NetBSD system.
One would then need six different library directories.  Once you have such
large systems, a generalised solution is obviously needed.

The existing solutions have been more thoroughly described in
\cite{tfheen-scalable}.  This is recommended reading to understand
some of the scalability problems in the software distribution systems
of today.

\section{The ``File Hierarchy Standard''}
\label{existing;fhs}

The current FHS\cite{fhs-site} only includes support for the
\directory{/lib}{\em qual} form with {\em qual} being 32 or 64 depending
on the pointer size.  There are proposals for adding other structures
with the most common being \directory{/lib/}{\em architecture}
\cite{taggart-multiarch} \cite{tfheen-multiarch3}.  Another proposal
was using the already existing support for a
\directory{/usr/}{\em architecture} which is well-supported in the
toolchain, but usually used for cross-compilation, not for binaries of
a ``foreign'' architecture.

\section{Current multiarch support in packaging tools}

The support for multiarch in the current set of packaging tools is
mostly missing with a few exceptions.  One such exception is RPM,
which supports installing packages for multiple architectures.  

By inspection of Fedora Core 3 RPMs, it does not seems to special-case
cross-architecture dependencies.  Most dependencies on
binary files are file-based dependencies rather than dependencies on
packages.  The dependencies are just on ``64 bit'' without any further
specification, so it is prone to confusing ia64-linux and
x86\_64-linux, for instance.  File dependencies by themselves are hard
to get right since you need to make sure only one package in the
distribution provides the file depended on.